Author's response to reviews

Title: Cerebral misery perfusion diagnosed using hypercapnic BOLD-contrast fMRI: a case report

Authors:

   Adam L Gordon (adam.gordon@nottingham.ac.uk)
   Stephen Goode (msxsg@nottingham.ac.uk)
   Olympio D’Souza (olympiodsouza@yahoo.com)
   Dorothee P Auer (dorothee.auer@nottingham.ac.uk)
   Sunil K Munshi (sunil.munshi@nuh.nhs.uk)

Version: 11 Date: 12 February 2010

Author's response to reviews:

We were asked to provide details of the research ethics approval for our research under the "Methods" section of the paper. By virtue of the journal's own framework, there is no "Methods" section, nor would it fit with the text of the article to list ethics approval within its syntax. We have therefore added a heading at the end of the article, after "Consent", entitled "Ethical Approval" which details that our research was approved by the Derby NHS Research Ethics Committee.